Author’s response to reviews

Title: Malignant perivascular epithelioid cell tumor of the lung synchronous with a primary adenocarcinoma: one case report and review of the literature

Authors:

Jikai Zhao (zhaojikai@126.com)
Haohua Teng (tenghh@live.cn)
Ruiying Zhao (ruiyingshao@foxmail.com)
Wenjie Ding (foxding0502@126.com)
Keke Yu (YKKXX@sina.com)
Lei Zhu (13621834637@163.com)
Jie Zhang (18017321572@163.com)
Yuchen Han (ychan@cmu.edu.cn;2303894233@qq.com)

Version: 2 Date: 23 Jan 2019

Author’s response to reviews:

1. Email addresses for all authors were included on the title page (Title page, line 9, page 1).

2. Separate headings ‘Discussion’ and ‘Conclusions’ had been combined into single ‘Discussion and Conclusions’ section (Abstract section, line 31, page 2).

3. We had amended the statement of Consent for publication in accordance with the requirements (Declarations section, line 193, page 13).

4. We had changed the authors’ contributions – initials in accordance with the requirements (Declarations section, line 203, page 13-14).
5. Zhu Lei’s name had been added onto the submission system, and everyone’s name is there.

6. Author Zhang Jie made a great contribution, we have added the description for this Author (Declarations section, line 205, page 13).

7. Contributions of authors Ruiying Zhao and Haohua Teng had been added (Declarations section, line 204, page 13).